Author's response to reviews

Title: Schwannoma of the external auditory canal: a case report

Authors:

Ozgul Topal (ozgultopal75@yahoo.com)
Selim S Erbek (selimerbek@gmail.com)
Seyra Erbek (seyraerbek@yahoo.com)

Version: 3 Date: 15 January 2007

Author's response to reviews:

Dear Editor,

we would like to submit the revised version of the manuscript. below are the details;

1. all unnecessary capitals were removed in the title and author names,

2. authors qualifications were removed

3. key words were removed from the manuscript

4. references were corrected

5. changed as figure legends

sincerely yours,
Ozgul Topal, MD